Author's response to reviews

Title: Using Patient Experiences to Improve Healthcare Quality: a multi-level study of Governmental Expectations and Hospital Practices in Norway

Authors:

Siri Wiig (siri.wiig@uis.no)
Karina Aase (karina.aase@uis.no)
Marianne Storm (marianne.storm@uis.no)
Martha Therese Gjestsen (mtgjestsen@hotmail.com)
Marit Solheim (marit.solheim@hisf.no)
Stig Harthug (stig.harthug@helse-bergen.no)
Glenn Robert (glenn.robert@kcl.ac.uk)
Naomi Fulop (n.fulop@ucl.ac.uk)

Version: 2 Date: 11 December 2012

Author's response to reviews:

As asked for by the journal editorial office, I have included the statement of the ethical approval and the reference number of the documents from the Norwegian Social Science Data Services, in the methods section.